Author’s response to reviews

Title: Systematic review of statistical approaches to quantify, or correct for, measurement error in a continuous exposure in nutritional epidemiology

Authors:

Derrick Bennett (derrick.bennett@ndph.ox.ac.uk)
Denise Landry (landry.denise@gmail.com)
Julian Little (jlittle@uottawa.ca)
Cosetta Minelli (cosetta.minelli1@imperial.ac.uk)

Version: 1 Date: 31 Aug 2017

Author’s response to reviews:

1. Please change the title of the 'Introduction' to 'Background'.

Done.

2. Please change the title of the 'Materials and Methods' section to just 'Methods'.

Done.

3. In the Authors' contributions section please add a statement confirming that all the authors read and approved the final version of the manuscript.

Done

4. Please provide a list of figure legends at the end of the main text (after the references).

Done.

5. Please remove the following two statements from the declarations section:

- Ethics: Not applicable
- Consent to Publish: All authors contributed to, read, and approved the final manuscript.

Done